Author’s response to reviews

Title: Curricular changes: The impact on medical students knowledge of neuroanatomy

Authors:

Mavilde Arantes (mavildearantes@hotmail.com)
José Paulo Andrade (jandrade@med.up.pt)
Joselina Barbosa (joselina@med.up.pt)
Maria Amélia Ferreira (mameliaferreira@med.up.pt)

Version: 1 Date: 09 Mar 2019

Author’s response to reviews:

Dear Esther Fagelson,

Many thanks for the helpful comments on the manuscript entitled "Curricular changes: The impact on medical students knowledge of neuroanatomy" which we submitted to BMC Medical Education.

Attached is a revised version in which we have attempted to address all issues raised in the referee comments. We agree with the suggestions and we have, therefore, changed the manuscript accordingly. We believe the manuscript is much improved as a result.